Author's response to reviews

Title: Childhood autism in a 13 year old boy with oculocutaneous albinism: a case report

Authors:

Muideen O Bakare (mobakare2000@yahoo.com)
Nkeiruka N Ikegwuonu (nkeiruikegwuonu@yahoo.com)

Version: 8 Date: 28 December 2007

Author's response to reviews: see over
The Editor
Journal of Medical Case Reports

Dear Sir,

RE: CHILDHOOD AUTISM IN A 13 YEAR OLD BOY WITH OCULOCUTANEOUS ALBINISM: A CASE REPORT

Many thanks for the comments of the Reviewers. Adjustments have been made to the manuscript based on the comments of the Reviewers.

Reviewer 1 (Suad Kapetanovic);

The Reviewer is satisfied with the present state of the manuscript and made no comment as regards further revision of the manuscript.

Reviewer 2 (Ann E. Maloney);

This Reviewer is also satisfied with the present state of the manuscript and made no comment as regards further revision of the manuscript.

Reviewer 3 (Karen Harman)

It has been stated and made clear under ‘Medical and Psychiatric History’ sub-heading that: “there was no history of any major medical illness that could influence normal neurological development in the patient during childhood and throughout the period of stay of the patient in the destitute home” as advised by the Reviewer.

Finally, the pictures had been made anonymous and downloaded.

Thank you.

Yours faithfully,

Dr. Muideen Owolabi Bakare, M.B.B.S, FMCPsych (Nigeria)

(Corresponding Author)